Author's response to reviews

Title: An open-label, sequential, dose-finding study of peginesatide for the maintenance treatment of anemia in chronic hemodialysis patients

Authors:

Anatole Besarab (abesara1@hfhs.org)
Steven N Zeig (pinesclinicalres@aol.com)
Edouard R Martin (edouard.martin@sfnamd.com)
Pablo E Pergola (ppergola@raparesearch.com)
Frederick C Whittier (clinicalresearch@sbcglobal.net)
Raja I Zabaneh (rz190@aol.com)
Brigitte Schiller (SchillerB@SatelliteHealth.com)
Martha Mayo (Martha_mayo@affymax.com)
Carol A Francisco (Carol_francisco@affymax.com)
Krishna R Polu (Krishna_polu@affymax.com)
Anne-Marie Duliege (Anne-marie_duliege@affymax.com)

Version: 3 Date: 17 August 2012

Author's response to reviews: see over
Friday, 17 August 2012

To Whom It May Concern:

On behalf of my coauthors, I would like to resubmit the manuscript “An open-label, sequential, dose-finding study of peginesatide for the maintenance treatment of anemia in chronic hemodialysis patients” for publication consideration in *BMC Nephrology*. Per instructions from the Assistant Editor (Dr. David Holmes), we have added the names of the ethics committees to the Methods section of the manuscript (pg 6). Also, we updated the manuscript per the formatting checklist.

If you have any questions regarding this resubmission, please do not hesitate to contact me via telephone (313-916-3711) or e-mail (abesaral@hfhs.org). Thank you for your consideration.

Sincerely,

Anatole Besarab, MD